Author's response to reviews

Title: ExaCT: automatic extraction of clinical trial characteristics from journal publications

Authors:

Svetlana Kiritchenko (svetlana.kiritchenko@nrc-cnrc.gc.ca)
Berry de Bruijn (berry.debruijn@nrc-cnrc.gc.ca)
Simona Carini (simona.carini@ucsf.edu)
Joel Martin (joel.martin@nrc-cnrc.gc.ca)
Ida Sim (sim@medicine.ucsf.edu)

Version: 7 Date: 24 September 2010

Author's response to reviews:

Dear Editor,

The requested formatting changes have been performed:
1. Highlighting is removed
2. Vertical lines in tables are removed
3. All figures are cropped
4. Consistent mention of figures is checked

Sincerely,
Dr. Svetlana Kiritchenko